Author's response to reviews

Title: Chronic musculoskeletal pain predicted hospitalisation due to serious medical conditions in a 10 year follow up study

Authors:

Hans Lindgren (hanslindgren@gmail.com)
Stefan Bergman (stefan.bergman@spenshult.se)

Version: 4 Date: 17 June 2010

Author's response to reviews:

The manuscript has now been changed according to requests from the editorial production team:
1. Affiliations with all information we have to give
2. Renumbered the tables and made the same changes in the text
3. Revised references so that all authors are given
4. Given the whole manuscript a last check

Kind regards
Stefan Bergman